Author's response to reviews

Title: Heat shock protein90 in lobular neoplasia of the breast

Authors:

Flora Zagouri (florazagouri@yahoo.co.uk)
Afrodite Nonni (afnonni@med.uoa.gr)
Theodoros N. Sergentanis (tsergentanis@yahoo.gr)
Nikolaos V. Michalopoulos (nmichal@med.uoa.gr)
Andreas C. Lazaris (alazaris@med.uoa.gr)
Efstratios Patsouris (epatsouris@med.uoa.gr)
George C. Zografos (gzografo@med.uoa.gr)

Version: 5 Date: 27 August 2008

Author's response to reviews:

Dear Editor, in the revised version of the manuscript I have completed the name of the Ethical Committee (typed in red in the revised manuscript), following your suggestion. In its current form I hope that it is suitable for publication to your distinguished journal.

Sincerely,

George C. Zografos